Author's response to reviews

Title: Overview of 71 European community based initiatives against childhood obesity starting between 2005 and 2011: general characteristics and reported effects

Authors:

Wanda JE Bemelmans (wanda.bemelmans@rivm.nl)
Trudy Wijnhoven (twi@euro.who.int)
Marieke Verschuuren (marieke.verschuuren@rivm.nl)
João Breda (JBR@euro.who.int)

Version: 2 Date: 21 October 2013

Author's response to reviews:

An explanation plus reference was added to the methods section of the manuscript related to the fact that no approval from an ethical committee was necessary for conducting the research described. We forgot to add this information to the original submission, for which we apologize.